Original Article



Measuring autism-associated traits in the general population: Factor structure and measurement invariance across sex and diagnosis status of the Social Communication Questionnaire Autism I–I5 © The Author(s) 2023

Article reuse guidelines: sagepub.com/journals-permissions DOI: 10.1177/13623613231219306 journals.sagepub.com/home/aut **S Sage**

Laura Hegemann^{1,2,3}, Ragna Bugge Askeland^{3,4}, Stian Barbo Valand^{1,2,3}, Anne-Siri Øyen², Synnve Schjølberg³, Vanessa H Bal⁵, Somer L Bishop⁶, Camilla Stoltenberg^{3,7}, Tilmann von Soest¹, Laurie J Hannigan^{2,3,8} and Alexandra Havdahl^{1,2,3}

Abstract

Autism screening questionnaires are sometimes used as a measure of "autism-associated traits" in samples drawn from the general population, even though such tools are primarily developed and designed for use in samples of children diagnosed with or being assessed for autism. Here, we explore the psychometric properties of the Social Communication Questionnaire (SCQ) current version reported at age 8 in a large population-based sample. Using data from the Norwegian Mother, Father and Child Cohort study (MoBa), we perform exploratory (N=21,775) and confirmatory (N=21,674) factor analyses on items and compare our results with previously suggested factor structure models of the SCQ. Furthermore, we test for measurement invariance across sex and registry-ascertained autism diagnoses, though with some qualitative differences in what the factors represent across these groups. This model performed largely consistently across boys and girls in the general population. Taken together, the SCQ's measurement properties must be carefully considered when it is used in population-based samples and measurement invariance testing of other autism screening tools used in similar contexts is warranted.

Lay abstract

Using questionnaires in research relies on the expectation that they measure the same things across different groups of individuals. If this is not true, then interpretations of results can be misleading when researchers compare responses across different groups of individuals or use in it a group that differs from that in which the questionnaire was developed. For the questionnaire we investigated, the Social Communication Questionnaire (SCQ), we found that parents of boys and girls responded to questionnaire items in largely the same way but that the SCQ measured traits and behaviors slightly differently depending on whether the children had autism. Based on these results, we concluded that researchers using this questionnaire should carefully consider these differences when deciding how to interpret findings. SCQ scores as a reflection of "autism-associated traits" in samples that are mostly or entirely made up of individuals without an autism diagnosis may be misleading and we encourage a more precise interpretation of scores as a broader indication of social-communicative and behavioral traits.

³Norwegian Institute of Public Health, Norway

⁷Department of Global Public health and Primary Care, University of Bergen, Norway

⁸MRC Integrative Epidemiology Unit (IEU), University of Bristol, UK

Corresponding author:

Laura Hegemann, Department of Psychology, University of Oslo, Oslo, Lovisenberg Diakonale Sykehus AS, PO Box 4970 Nydalen, NO-0440, Norway.

Email: laura.hegemann@lds.no

¹Department of Psychology, University of Oslo, Norway

²Nic Waals Institute, Lovisenberg Diaconal Hospital, Norway

⁴Oslo University Hospital, Norway

⁵Graduate School of Applied and Professional Psychology, Rutgers University, USA

⁶Department of Psychiatry and Behavioral Sciences, University of California San Francisco, USA

Keywords

cohort studies, factor analysis, MBRN, measurement invariance, MoBa, psychometrics, statistical, surveys and questionnaires

Introduction

Social communication difficulties and restricted, repetitive patterns of behavior or interests (RRBI) are the core diagnostic criteria for an autism spectrum diagnosis (hereafter autism) but are also multidimensional areas of development and behavior with relevance across the wider population. One group in which higher rates of these traits are typically seen is in non-autistic family members of those with a diagnosis (Frazier et al., 2015; Piven et al., 1997). This observation has had broad implications for etiological understandings of autism, including contributing to dimensional views of autism. However, social communication difficulties and RRBI are non-specific to autism (Lord & Bishop, 2021). Deriving insights about autism from studies of these traits in non-autistic or predominantly nonautistic samples therefore relies on the assumption that questionnaire-based screening instruments primarily developed and tested in samples of individuals with an autism diagnosis measure the same thing in populationbased and largely non-autistic samples. Verifying this assumption-known as "measurement invariance"-is necessary to clarify how information from populationbased studies of social communication difficulties and RRBI can be interpreted and used.

There are several questionnaire-based screening instruments for autism, measuring a range of social communication difficulties and RRBI, and developed for use in different contexts and ages. The Social Communication Questionnaire (SCQ; Rutter et al., 2003), designed to parallel the Autism Diagnostic Interview-Revised (Lord et al., 1994), is a parental report screener for children aged 4 years and older. The questionnaire covers three domains reflecting Diagnostic and Statistical Manual of Mental Disorders (4th ed.; DSM-IV) criteria of social interaction, communication, and RRBI. The SCQ was originally intended either for use as a second-level screening instrument (i.e. to be administered after initial concerns about development have been raised) or as a comparative measure of signs of autism between clinical groups or over time (Rutter et al., 2003). However, the SCQ, as well as other autism questionnaires, are sometimes used in general population or population-based samples and other largely nonautistic samples for a range of different applications, including as a quantitative measure of "autism-associated traits" (Bjørk et al., 2018; Firestein et al., 2022; Kim et al., 2021; Sasson & Bottema-Beutel, 2022).

There is substantial literature on the screening accuracy of the SCQ for identifying children with autism. Most of these studies are in clinical samples, but there are some in

population-based samples (Chesnut et al., 2017; Surén, Saasen-Havdahl, et al., 2019) and a previous study has examined distributions of SCQ-scores in the general population (Mulligan et al., 2009). However, there is limited work assessing what trait dimensions or factors are measured by the SCQ in the general population and whether these factors differ or are measured differently between those with and without autism. The current version of the questionnaire as opposed to the life-time version may be more relevant in analyses using the SCQ as a measure of continuous traits given it measures behaviors present at the time of administration. Therefore, investigating the psychometric proprieties of the SCQ current version could be informative to the appropriate interpretation and use of the instrument as an "autism-associated trait" measure in population-based samples.

The factor structure of the SCQ, for both the life-time and the current versions of the instrument, has been investigated in primarily clinical samples (Berument et al., 1999; Gau et al., 2011; Grove et al., 2019; Karaminis & Stavrakaki, 2022; Kidd et al., 2020; Liu et al., 2022; Magyar et al., 2012; Martin et al., 2014; Wei et al., 2015) and in several samples when combined with other autism or attention-deficit hyperactivity disorder (ADHD) questionnaires (Krakowski et al., 2022; Warrier et al., 2022). Factor analysis aims to explain the correlational structure among items in a scale by modeling latent factors that are thought to influence individuals' responses on the items. Factor analyses of the SCQ have typically tested the fits of models based on clinical criteria or ascertained in a clinical cohort. Commonly these studies include either or both a 2-factor model based on the Diagnostic and Statistical Manual of Mental Disorders (5th ed.; DSM-5) criteria for autism (social communication and RRBI), a three-factor model based on DSM-IV criteria (social interaction, communication, and RRBI) and/or the Berument et al. (1999) solution containing factors covering social behavior, communication, atypical language, and stereotyped behavior. Less is known about the factor structure in general population samples, and prior work has relied on using models based on theories developed in autism samples, such as diagnostic criteria (Evans et al., 2019), and theorizing them to be translatable to general population samples.

In addition to questions surrounding measurement consistency across diagnosis status, it is also necessary to consider potential sex differences in the properties of autism instruments. Understanding whether and how the SCQ performs differentially across sex is important as the underpinnings of sex differences in autism is an active area of research. For example, the "female protective effect" hypothesis suggests that females require a higher magnitude of predisposing factors (usually genetic liability) to be diagnosed with autism compared to males (Robinson et al., 2013; Wigdor et al., 2022). In addition, diagnostic considerations reflecting biases due to less recognition of differences in female presentations of autism or higher compensatory behaviors (Lai et al., 2015, 2017; Livingston & Happé, 2017) may lead to sex differences in the psychometric properties of autism instruments. The implications of these explanations differ markedly, so research to help disentangle them is vital. Data from instruments such as the SCQ applied in the general population may be able to contribute to this research, but understanding how their measurement differs (or not) across sex is necessary to make valid inferences.

A few studies have addressed sex differences of the SCO in primarily autistic samples (Evans et al., 2019; Karaminis & Stavrakaki, 2022; Krakowski et al., 2022; Wei et al., 2015). These studies find some differences in specific items' psychometric properties such as item functioning or how often items are endorsed. Krakowski et al. (2022) tested formally for measurement invariance; however, in these analyses, the SCO was combined with an ADHD measure. Sex differences in the performance of the SCQ, and other instruments like it, may be difficult to detect in clinical samples as these samples are more likely to be predominantly male and might already reflect a sex bias in selection and ascertainment. Population-based samples in which parents enroll on behalf of their children do not usually suffer from this kind of ascertainment bias relating to the sex of the child.

In the present study, we take advantage of a large-scale population-based sample, the Norwegian Mother, Father, and Child Cohort Study (MoBa) cohort, with linkage to health registry data containing information about autism diagnoses to: (1) take a hypothesis-free approach to examine the underlying structure of the SCQ current version in a population-based sample at age 8; (2) compare this structure to factor structures derived from clinically ascertained samples or based on diagnostic criteria; and (3) assess assumptions of measurement invariance across both sex and autism diagnostic status.

Methods

Sample

The Norwegian Mother, Father and Child Cohort Study (MoBa) is a prospective population-based pregnancy cohort study conducted by the Norwegian Institute of Public Health (Magnus et al., 2006, 2016). Participants were recruited from all over Norway from 1999 to 2008. The women consented to participation in 41% of the

pregnancies. The cohort includes approximately 114,500 children, 95,200 mothers, and 75,200 fathers. The current study is based on version 12 of the quality-assured data files released for research in January 2019. The establishment of MoBa and initial data collection was based on a license from the Norwegian Data Protection Agency and approval from The Regional Committees for Medical and Health Research Ethics. The MoBa cohort is currently regulated by the Norwegian Health Registry Act. The current study was approved by The Regional Committees for Medical and Health Research Ethics (2016/1702).

We use data from all MoBa children for whom mothers had completed the SCQ (Rutter et al., 2003) when their children were 8 years old (N=43,449; $M_{age}=8.18$, SD=0.18 years). Children's sex was ascertained from the Medical Birth Registry of Norway (MBRN) so that sex refers to sex assigned at birth rather than gender. In total, 51% of the children in the sample were male and 49% female. Highly educated and older mothers are somewhat overrepresented in MoBa. Furthermore, participants with lower education levels were lost to follow-up at higher rates than those with higher education levels, increasing the overrepresentation of highly educated mothers in the present sample (Biele et al., 2019; Vejrup et al., 2021). Mothers of children with an autism diagnosis had lower completion rates of the 8-year questionnaire than mothers without (~5% difference) but similar rates of missingness across items within the SCQ. Children who were reported to not have phrase speech at the time of completing the questionnaire were a small minority in our sample, only making up 2.8% (N = 18) of those with an autism diagnosis and 0.7% (N = 294) of those without. Participants were not asked to report ethnicity.

Measures

We included 38 of the 40 dichotomous (yes/no) items from the current version of the SCQ. Item 1, "Is your child able to talk using short phrases or sentences," was excluded from factor analyses because it is not included in the scoring but only intended to assess whether the verbal items (Items 2–7) are applicable. Those who answered "No" to Item 1 were set to missing for those items. We also excluded Item 25, "does your child shake his/her head to indicate no," as it was almost perfectly correlated with Item 24 "does your child nod his/her head to indicate yes."

Clinical diagnoses of autism in MoBa participants were captured by the Norwegian Patient Registry (NPR) between 2008 and June 2021. The subsample of individuals with a recorded autism diagnosis included International Classification of Diseases, 10th Revision (ICD-10) codes for F84.0 (Childhood Autism), F84.1 (Atypical Autism), F84.5 (Asperger syndrome), F84.8 (Other pervasive developmental disorder), and F84.9 (Pervasive developmental disorder unspecified).

Exploratory and confirmatory factor analyses

We conducted exploratory factor analyses (EFA) in one randomly selected half of the sample (EFA-sample; N=21,775). To determine the optimal number of factors to retain, we used a scree plot; results from parallel analysis, optimal coordinates, and acceleration factor; as well as compared fit indices (comparative fit index (CFI), Tucker-Lewis index (TLI), standardized root mean square residual (SRMR), root mean square error of approximation (RMSEA)) across models extracting 1-10 factors. Theoretical interpretability of the factors and parsimony of the model were also considered when selecting the best solution. Based on these metrics, models with reasonable performance were run using confirmatory factor analysis (CFA) using the other half of the sample (CFA-sample; N=21,674). In these models, items were specified to only load onto their highest loading factor in the EFA. The final model was selected from these models balancing fit indices from the CFA with parsimony of the model. As a sensitivity analysis, we also ran the final model based on the EFA results with a method factor to account for negatively worded items. In the CFA sample, three alternative factor models based on previous literature were also run. The alternative models were: a 2-factor model with two factors covering the DSM-5 dimensions; a 3-factor model with the SCQ domains based on DSM-IV criteria; and the Berument et al. (1999) 4-factor solution.

Measurement invariance testing

Prior to formal measurement invariance testing, four models (the final model derived from the EFA and the alternative models) were run in subsamples of our total sample based on sex (N_{males} =22,119, $N_{females}$ =21,253) and autism diagnosis status (N_{nodx} =42,736, N_{dx} =636), with sensitivity analyses excluding individuals that were present in the sample used for the EFA. Measurement invariance testing was then conducted across both sex and autism diagnostic status by means of multiple group structural equation modeling, where groups were defined according to sex and having received a diagnosis of autism or not, respectively. We tested increasingly more restrictive assumptions¹ on which parameters of the model can be assumed to be equal across groups. Specifically, we first tested for configural *invariance* by examining whether the factor structure of the SCQ could be replicated across groups. Second, we tested for equality of factor loadings and thresholds across groups. Finally, we tested for the most restrictive assumption by testing the equality of the residual variances across groups (Kline, 2016; Wu & Estabrook, 2016).

We used change in the CFI and McDonald's Noncentrality Index (McNCI) for formal comparison across the models. In the autism diagnosis status models, to account for the highly unbalanced group sample sizes, we ran 100 iterations of the models randomly subsampled with replacement from the non-diagnosed group. We used an average of the change in fit indices across the models to make the comparisons (Yoon & Lai, 2018). To assess if measurement invariance was met (i.e. whether constraints on parameters across groups at each level were acceptable), we used both conservative criteria (CFI: $-\Delta 0.002$ and McNCI: $-\Delta 0.008$; Meade et al., 2008) as a lower bound and standard criterion (CFI: $-\Delta 0.01$ and McNCI: $-\Delta 0.02$; Cheung & Rensvold, 2002) as a upper bound to reject the hypothesis of invariance, with results falling between being considered inconclusive and interpreted alongside differences in parameter estimates. For full technical details of the measurement invariance testing, sensitivity models testing partial invariance, the selection of fit indices for model comparisons, and the subsampling approaches for the diagnostic status analysis, see the supplementary methods.

Software and analytic code

EFA and CFA were conducted using version 8 of the statistical software *Mplus* (Muthén & Muthén, 1998–2017). Analyses to estimate the number of factors to extract and measurement invariance models were run using the *nFactors* (v2.4.1), *psych* (v2.1.9), *lavaan* (v0.6-14), and *sem-Tools* (v0.5-6) packages in R version 4.1.2 (Jorgensen et al., 2022; Raiche & Magis, 2020; R Core Team, 2021; Revelle, 2021; Rosseel, 2012). Diagnostic data was ascertained using the MoBa *phenotools* (v2.9) package (Hannigan et al., 2023). See supplementary methods for further details.

Analytic code for the analyses is openly available at https://github.com/psychgen/autism-scq-meas-invariance. The consent given by the participants does not allow for storage of data on an individual level in repositories. Researchers can apply for access to data for replication purposes via MoBa, in line with MoBa data access policies.

Community involvement statement

The general population of parents and youth in MoBa have been engaged in focus groups guiding the development of the MoBa questionnaires. The Norwegian national associations for autism have provided input on selection of measures to include in the MoBa questionnaires, including the SCQ. Discussions and ultimate decision to include the SCQ in MoBa were by the Autism Birth Cohort Study (Stoltenberg et al., 2010). The Lovisenberg Hospital User Panel provided input on the development of the grant application funding this work and discussion of results.

Results

The mean total score for the SCQ was 3.22 (SD=2.81) in the full sample and 3.65 (SD=3.03) in boys, 2.76 (SD=2.48) in girls, 7.99 (SD=6.01) in those with an autism diagnosis,



Figure 1. (a) Scree plot and results of the parallel analysis, eignvalues, optimal coordinates, and acceleration factor are present. (b) Fit indices for EFAs extracting I–10 factors. Fit criteria; RMSEA: excellent < 0.01, good < 0.05, mediocre < 0.08, < 1 poor. CFI and TLI: excellent > 0.97, good > 0.95, acceptable > 0.9, poor < 0.9. SRMR: acceptable < 0.10, good < 0.08.

and 3.15 (SD=2.67) in those without. Scores for subdomains (Table S1) and item prevalence rates (Tables S2–S6) in the full sample and across sex and diagnosis status alongside formal comparisons of score means and item prevalence across the groups (Tables S1, S7) are presented in the supplementary materials.

Ascertaining the factor structure of the SCQ in a general population sample

Exploratory factor analyses. Results from EFA of the SCQ are presented in Figure 1. Statistical methods and interpretation of the scree plot (panel a) alongside fit indices (panel b) indicated solutions with 2–7 factors. Ultimately, the 5-factor solution was selected because it was a more parsimonious model than the 6- and 7-factor solutions and, compared to the 2- to 4-factor models, had better fits in both the EFA and CFA samples (EFA: panel b; CFA: Table 1) with more interpretable factors.

The five identified factors in the EFA, indicated by the darkened bars in Figure 2, correspond to the domains of idiosyncratic speech (e.g. "repeat things in exactly the same way"), repetitive and restrictive behaviors or interests (e.g. "interests that might seem odd"), social reciprocity (e.g. "smiles back if someone smiles at them"), non-verbal communication (e.g. "uses gestures other than pointing/pulling"), and play (e.g. "plays imaginative games with another child"). Most items were fair indicators of their respective factors ($\lambda > 0.4$) with four exceptions (Items 9, 17, 19, and 34; see Figure 2). Only Items 2 and 28 had cross loadings greater than 0.4, both being under 0.45. Solutions with fewer factors had substantially more items with cross loadings greater than 0.4 and/or weaker estimated loadings on their primary factor than in the 5-factor model (Tables S8–S11).

Confirmatory factor analyses. In the CFA using the other half of the sample, the 5-factor model (Figure S1)



-	-				
Model	Chi-square	CFI	TLI	RMSEA	SRMR
Models based on results from the EFA					
3-factor model	χ ² (662) = I 2,308.707	0.803	0.790	0.028	0.116
4-factor model	$\chi^2(659) = 10,132.608$	0.839	0.829	0.026	0.111
5-factor model	$\chi^2(655) = 6437.532$	0.902	0.895	0.020	0.087
5-factor model w/ method factor	$\chi^2(640) = 13,169.286$	0.894	0.884	0.021	0.081
Alternative models					
2-factor (DSM-5 criteria)	χ ² (593) = 16,028.627	0.707	0.689	0.035	0.153
3-factor (SCQ Domains/DSM-IV)	$\chi^2(557) = 22,724.532$	0.612	0.586	0.042	0.156
4-factor (Berument et al., 1999)	$\chi^2(659) = 15,324.240$	0.751	0.735	0.032	0.143

Table 1. Fit indices presented for models run in the CFA sample.

CFA: confirmatory factor analysis; CFI: comparative fit index; TLI: Tucker–Lewis index; RMSEA: root mean square error of approximation; SRMR: standardized root mean square residual; EFA: exploratory factor analyses; DSM-5: Diagnostic and Statistical Manual of Mental Disorders (5th ed.); SCQ: Social Communication Questionnaire; DSM-IV: Diagnostic and Statistical Manual of Mental Disorders (4th ed.).

as developed through the EFA fits the data adequately according to most fit indices, even though the TLI was somewhat low (Table 1). The addition of a question wording method factor decreased fit slightly for most indices. In this half of the sample, the items in the 5-factor model were good indicators of their respective factors as only two items had standardized factor loadings under 0.4 (Items 9 and 19). Factors were positively correlated with each other except for the "non-verbal communication" factor which was slightly negatively correlated with "restrictive, repetitive behaviors or interests" and "idiosyncratic speech" factors, and the "play" factor which was uncorrelated with the "idiosyncratic speech" factor.

We then compared the 5-factor model to alternative factor structures for the SCQ from the literature (Table 1). The Berument 4-factor model fit the data poorly but was the best out of the alternative models tested. The 2-factor DSM-5 model and the original 3-factor SCQ factor model also fit exceptionally poorly. Items 9 and 19, which were poor indicators in the 5-factor model, also had factor loadings under 0.4 for their specified factors in any of the alternative factor models. The idiosyncratic speech factor identified in the 5-factor solution was the same as the atypical language from the Berument 4-factor model. Other factors did not directly overlap with factors from other solutions, but the RRBI factor was very similar across the 5-factor solution, 3-factor DSM-IV model, and Berument model. The non-verbal communication and social reciprocity factors were closest to the social and communication factors in the 3-factor DSM-IV model.

Testing measurement invariance across sex and diagnostic status

Sex. The 5-factor model provided an overall acceptable fit to the data when run separately in boys and girls, although CFI/TLI were slightly under an acceptable range in boys (Girls—CFI: 0.911, TLI: 0.905, RMSEA: 0.017, SRMR: 0.091; Boys—CFI: 0.891, TLI: 0.883, RMSEA: 0.022,

SRMR: 0.083). This model was a better fit in both groups than the alternative models from the literature (Table S16). The estimated factor correlations from these models are shown in Figure 3, with girls below the diagonal and boys above. Factors were similarly correlated in both groups.

We found that both the configural invariance models had acceptable fits comparable to the model in the full sample (Table 2) and invariance of thresholds and loadings held, indicating the same factor structures for boys and girls, with factors measuring equivalent constructs. When assessing equivalent residual variances of the items, the invariance assumption was inconclusive. Further investigations revealed that the differences in fit indices could be attenuated by freeing the residual variance of Item 13 ("special interests of unusual intensity"), indicating that sex differences in measurement precision could be largely attributed to this item.

Diagnostic status. The 5-factor model fit well when run separately in those with and without a registered autism diagnosis (Dx-CFI: 0.929, TLI: 0.924, RMSEA: 0.031, SRMR: 0.095; NoDx-CFI: 0.914, TLI: 0.907, RMSEA: 0.019, SRMR: 0.084) even when excluding individuals included in the sample for the EFA to guard against overfitting (Table S16). The 5-factor model had the best fit compared with the alternative models from the literature in both groups, although the alternative models had relatively better fits in the diagnosis group than in the no diagnosis group (Table S16). The factor correlations from the 5-factor model in the two groups are shown in Figure 3, with the no diagnosis group below the diagonal and the diagnosis group above. The estimated correlations differed slightly across diagnostic status as, broadly speaking, the factors were more correlated in the diagnosed group than in the no diagnosis group. Particularly, the "repetitive and restrictive behaviors" and, to a lesser extent, the "idiosyncratic speech" factors were more correlated with factors covering social and communication traits in the diagnosed group.



Figure 3. Correlations matrices of the 5-factor model when run in subsamples for (a) males (above the diagonal) and females (below the diagonal) and (b) those with a registered diagnosis code of autism in NPR (above) and without (below).

Model	Compared to	CFI	∆CFI	McNCI	$\Delta McNCI$	TLI	RMSEA	Invariance holds
Sex								
Configural	NA	0.902		0.862		0.895	0.020	Yes
Loadings/Thresholds	Configural	0.906	0.004	0.859	-0.003	0.901	0.019	Yes
Residual variances	Loadings/Thresholds	0.901	-0.005	0.847	-0.012	0.899	0.019	Yes/No
Autism diagnostic status								
Configural	NA	0.914		0.838		0.907	0.019	Yes
Loadings/Thresholds	Configural	0.909	-0.005	0.816	-0.022	0.904	0.020	Yes/No
Residual variances	Loadings/Thresholds	0.905	-0.003	0.789	-0.027	0.903	0.020	Yes/No

Table 2. Measurement invariance testing across sex.

CFI: comparative fit index; McNCI: McDonald's Noncentrality Index; TLI: Tucker–Lewis index; RMSEA: root mean square error of approximation; NA: not applicable.

Conservative Criteria; $\Delta CFI = -0.002$, $\Delta McNCI = -0.008$. Standard criteria: $\Delta CFI = -0.01$, $\Delta McNCI = -0.02$. Yes/No refers to being under the standard criteria for either one or both ΔCFI and $\Delta McNCI$ but exceeding the conservative criteria in both. Fit indices for the autism diagnostic status models represent the averaged indices over 100 iterations of the models, 99 of which converged.

The configural model across diagnosis status had an adequate fit (Table 2). Invariance of the thresholds and loadings could not be conclusively confirmed or rejected, as Δ CFI fell in-between our conservative and standard criteria we set to conclusively reject the hypothesis of invariance while Δ McNCI exceeded both criteria. Examining parameter estimates to gain insight on the why change in both fit indices exceed our conservative fit criteria, we note that estimated loadings (Figure 4) and thresholds for some items between groups differed in the configural model. Some of these differences were sizable; for example, one

loading showed a 66% decrease in magnitude from the diagnosed to the no diagnosis group. For comparison, between males and females—where fit indices conclusively supported the invariance of thresholds and loadings—the largest percent difference in loading between groups was 22%. As invariance of the thresholds and loadings held based on standard Δ CFI criteria, the invariance model testing equivalence of the residual variances was run but invariance assumption failed to hold based on Δ McNCI and was inconclusive based on Δ CFI. Due to the invariance of thresholds and loadings not holding based on Δ McNCI,





we did not perform significance testing on differences seen in the factor correlations as the factors may represent different constructs and thus not directly comparable. Altogether, results indicated that the factor structure was the same across the two groups, but the factors may represent somewhat different constructs in those with and without an autism diagnosis.

Discussion

A 5-factor model best accounted for the observed patterns of relationships among the SCQ items rather than solutions with fewer factors, suggesting higher dimensionality underlying these traits in a population-based cohort than, for example, two factors of social/communication and RRBI that may be assumed based on the current diagnostic criteria for autism. This model was the best fitting model in both boys/girls and those with and without an autism diagnosis. Formal measurement invariance testing supported similar factor structures across both sex and autism diagnosis status as well. More restrictive measurement invariance tests, however, suggested that while factors represented the same constructs between males and females, the nature of what the identified factors measure may differ between those with and without autism diagnoses.

Factor structure underlying the SCQ in the general population

These results provide an important reminder that the factor structure of autism screening instruments in populationbased samples cannot be assumed to be the same as those derived from clinical populations-especially when such samples clearly differ from those used in the development or testing of the instrument. Our results indicate that the SCQ measures social, communication, and RRBI traits in a population-based cohort across five dimensions. This is notably more than the 2-3 domains based on DSM-IV or DSM-5 criteria that is typically seen. The 5-factor model performed markedly better in our population-based sample than the alternative factor structures based on theory developed from autism clinical samples. This is consistent with a similar finding of increased heterogenicity underlying another frequently used autism screening instrument in the general population (Taylor et al., 2020). While the 5-factor model also provided the best fit to the data in the subsample of individuals with an autism diagnosis, the 2- to 3-factor models previously identified in clinical samples were relatively better fits in this group compared to the full sample.

In our 5-factor solution, the RRBI factor, the social reciprocity factor, and the non-verbal communication factor were similar to the corresponding factors from the 3-factor model based on *DSM*-IV criteria. The additional factors, both with comparably fewer items loading onto them, we labeled idiosyncratic speech and play. While

items in the idiosyncratic speech factor have separated from larger factors in previous EFA work in clinical samples, including in the Berument solution (Berument et al., 1999; Warrier et al., 2022), the items in the play factor have not. This may reflect that many things influence differences in play behaviors, leading to reduced correlations of the items with the other social factors. Therefore, we interpret this finding as evidence of more complex heterogeneity in play behaviors in the general population. Practically, this suggests that sum scores with these items, particularly in a population-based cohort, are likely capturing multiple constructs which should inform interpretation of associations with these scores used in this context.

We found poor loadings of several more items, most covering social and communication traits, indicating some further heterogeneity influencing responses factor. Item 19 ("has a particular best friend") did not have strong loadings on any factor—either in our 5-factor model in both the EFA and CFA or in any of the alternative models. This was also the only item that was a universally poor indicator in our study. The item just barely crossed a 0.4 threshold in those with a diagnosis and was a poor indicator in both girls and boys. The other items with low loadings in either our general population EFA or CFA models (EFA: Items 9, 17, 34; CFA: Item 9) have been observed as good indicators in clinical populations (Berument et al., 1999; Grove et al., 2019) and performed well in many of the models in our diagnosed subsample.

Besides the items with low loadings in the general sample, there were several other observations from our results that may suggest further heterogeneity being captured by the SCO items in the general population: (1) many items had significant, although generally small, cross loadings across multiple factors in the EFA indicating items are measuring multiple constructs; (2) none of the models tested, either developed from our EFA or based on our literature, had more than just "acceptable" fits using standard CFI, TLI, and SRMR criteria in the general population; and (3) several methods to determine the number of factors indicated even more underlying factors. These observations may also highlight difficulties in using the SCQ as "trait" measure of underlying factors given its binary response options and intended use as a screener. Taken together, we suggest sum scores of the SCQ, particularly total scores or subscores based on DSM-5 diagnostic criteria, are likely capturing multiple factors. This supports arguments against using a total score for distinguishable and uncorrelated characteristics which would lack the coherent meaning implied by the term "autism traits" (Bottema-Beutel et al., 2021). Therefore, we recommend against using a SCQ total score as a measure of autistic traits in future etiological work in population-based samples. Instead, we note examining subscale or even itemlevel differences may be informative depending on the empirical context.

Measurement invariance across sex

We found no evidence of any major sex differences in the way the SCQ measures social communication and RRBI traits in the general population. This is somewhat surprising given the literature on differences in presentations of autism behaviors in girls (or even a female autism phenotype) and concerns that current instruments may not be adequately capturing these behaviors in girls (Hull et al., 2020; Mandy & Lai, 2017). In a population-based sample, the SCQ can be assumed to be measuring the same underlying factors across sexes. The strictest invariance model failed to meet conservative criteria, suggesting that there may be small differences in the precision of the instrument to measure the factors between males and females.

The potential differences in measurement precision seems to be accounted for by Item 13 ("special interests of unusual intensity") as partial invariance of the residual variances holds when the parameter for the item was allowed to differ between groups. This concurs with previous studies that have found this item to have differing psychometric properties between sexes, such as significant differential item functioning between girls and boys in an item response theory framework (Wei et al., 2015) and higher in the psychometric property of difficulty in girls compared to boys (Karaminis & Stavrakaki, 2022). The changes in fit indices between the invariance models are small and using standard criteria would support invariance of the residual variance. This suggests that, overall, observed scores between boys and girls in a populationbased sample can be directly compared. However, given agreement with the previous literature regarding sex differences in the psychometric properties of Item 13, we caution against interpretation of sex differences in endorsement of this item as solely due to differences in the underlying behavior.

Measurement invariance across autism diagnostic status

The 5-factor model provided a good fit in both those with and without autism, and configural invariance testing indicated that the number of factors and items loading onto those factors can be assumed to be the same across the two groups in our sample. Differences between the two groups emerged, however, when we tried to constrain items to be of equal importance to their factors across groups. Based on formal measurement invariance testing, we failed to conclusively uphold or reject the invariance of thresholds and loadings as both alternative fit indices exceed our conservative criteria to reject invariance but Δ CFI did not exceed the standard criteria, making the formal comparison inconclusive. Considering qualitative observations, some items had noticeable differences in estimated standardized factor loadings between the two groups. In some cases, these differences were observed to be systematic across factors (e.g. consistently higher loadings for RRBI and social reciprocity items among the autism group). Taking together, we think there is enough uncertainty in the results to prevent us from concluding that the relationship between observed scores and the factors are definitely the same between the two groups. Thus, we recommend caution should be taken in interpreting identical observed scores across two individuals with and without an autism diagnosis as equality of the individuals on the underlying factor. Parental concern about autism has been shown to influence parental reporting on other autism measures (Havdahl et al., 2017) and likewise these findings may be reflecting differences in mothers' interpretation of the items. For example, parents with children with autism, who are familiar with the behaviors intended to be measured by SCO items, may interpret the items differently than parents of non-autistic children who may be thinking of different behaviors. This should be considered in the use of SCQ scores in general population samples, as it emphasizes that the SCQ is more appropriately used as a measure of social communication or RRBI rather than "autismassociated traits."

Limitations

There are several limitations that should be kept in mind in the interpretation of these results. First, despite our efforts to minimize over-fitting-for example, by splitting our sample in half for the EFA and CFA-all analyses were exploratory and data-driven. Moreover, the relatedness between participants in MoBa (both in terms of siblings and more structural relatedness owing to MoBa parents being related to other MoBa parents) meant that even splitting the sample at random did not completely remove dependency between the two halves. Notwithstanding familial relatedness between the two half samples, other similarities (due to them coming from the same study, and therefore having been ascertained in the same nonrandom way) also limit their independence and increase the risks of over-fitting. Taken together, this means that over-fitting likely made some contribution to our results. While overfitting can result in spuriously complex models (Forster & Sober, 1994), it is highly unlikely to be solely responsible for the large differences in fit between our 5-factor model and the simpler alternative structures we tested in the general population. Beyond simply the scale of these differences, the solution we identify is theoretically coherent, with 4 out of the 5 identified factors similar to those seen in previous factor solutions, and the separation of an additional "play" factor reflecting diverse contributions to difficulties in play behaviors in a population-based sample.

Second, autism diagnoses were ascertained by ever receiving a diagnostic code for autism in the national public registry from the years of 2008–2021. As a result, there may be individuals in the "no diagnosis" group who received diagnoses prior to 2008/after 2021 or remain undiagnosed. Misdiagnosis in the autism group is also a possibility, although 93% of those receiving a diagnostic code once in our sample received it a second time. A record review of MoBa data conducted in previous work showed few coding errors and that most autism diagnoses were well documented in health records (Surén, Havdahl, et al., 2019). In addition, the autism diagnosis group may differ from other clinical cohorts due to selective participation among MoBa families. Relatively few in the group were reported to not have phrase speech or co-occurring intellectual disability and had an older average age of first recorded diagnosis (9.15 years for those with NPR data available from birth, meaning a portion were not diagnosed when mothers filled out the questionnaire). These factors likely reflect a group that differs in make-up from what is typically seen in clinically ascertained cohorts, potentially contributing to the better fit of the five-factor model in the autism subsample compared to the alternative models from the literature. Furthermore, measurement invariance testing based on heterogeneous aspects of autism such as language ability may be beneficial. Finally, due to the low number of autism diagnoses in females in our sample and issues with model estimation relating from this, we chose not to compare model structures and measurement invariance by diagnostic status and sex, limiting what we can conclude on sex differences in measurement between general and clinical populations.

Conclusion

Autism questionnaire-based screening instruments, such as the SCO, can be used as a measure of social communication and RRBI traits in general population samples, but their relevance to understanding the etiology of autism depends on their measurement properties in these samples. The evaluation of the factor structure in the present analyses suggests a more nuanced multidimensionality among these traits in population-based samples-more so than is implied by structures previously derived from data collected in clinically ascertained samples. While we find the instrument's measurement to be largely invariant across sex, we find some evidence that the dimensions measured by the SCQ—while consistent in terms of structure—may differ qualitatively between autistic and non-autistic individuals. Taken together, we caution against assuming endorsement of SCQ items as autism-associated traits in population-based samples. Furthermore, in future research using SCQ scores in population-based samples, using more than just total scores may be informative. In addition, care must be taken not to extend conclusions to the etiology of autism specifically, but rather the etiology of social, communication, and RRBI traits more generally. Finally, our findings add support to the importance of looking at

measurement properties of other autism screening instruments used as trait measures in population-based samples.

Acknowledgements

The Norwegian Mother, Father and Child Cohort Study is supported by the Norwegian Ministry of Health and Care Services and the Ministry of Education and Research. We are grateful to all the participating families in Norway who take part in this ongoing cohort study. The Autism Birth Cohort Study provided the funding for the repetitive linkage between MoBa and NPR. This work was performed on the TSD (Tjeneste for Sensitive Data) facilities, owned by the University of Oslo, operated and developed by the TSD service group at the University of Oslo, IT Department (USIT) (tsd-drift@usit.uio.no). Measurement invariance testing was performed on resources provided by Sigma2 the National Infrastructure for High-Performance Computing and Data Storage in Norway.

Declaration of conflicting interests

The author(s) declared the following potential conflicts of interest with respect to the research, authorship, and/or publication of this article: V.H.B. receives research funding from Western Psychological Services for research related to the Autism Diagnostic Interview–Revised and receives consultant fees from Janssen Research and Development, Inc.

Funding

The author(s) disclosed receipt of the following financial support for the research, authorship, and/or publication of this article: The South-Eastern Norway Regional Health Authority supported L.H. (#2020022), A.H. (#2020022, #2018059), S.B.V. (#2022029), and L.J.H. (#2019097, #2022083, #2018058). R.B.A. was supported by the Research Council of Norway (#274611).

Disclaimer

Data from the Norwegian Patient Registry has been used in this publication. The interpretation and reporting of these data are the sole responsibility of the authors, and no endorsement by the Norwegian Patient Registry is intended nor should be inferred.

ORCID iDs

Laura Hegemann D https://orcid.org/0000-0002-7398-3760 Ragna Bugge Askeland D https://orcid.org/0000-0002-7465-0634 Vanessa H Bal D https://orcid.org/0000-0003-0750-823X

Supplemental material

Supplemental material for this article is available online.

Note

 Due to Social Communication Questionnaire (SCQ) items having dichotomous response sets, there are some limits on which invariance assumptions that can be tested independently; we formally tested three assumptions following recommendations from Wu and Estabrook (2016) and therefore deviate from using the terms weak/metric, strong/scalar, and strict invariance in favor of invariance of the specific parameters we examine. See supplementary methods for further details.

References

- Berument, S. K., Rutter, M., Lord, C., Pickles, A., & Bailey, A. (1999). Autism screening questionnaire: Diagnostic validity. *The British Journal of Psychiatry*, 175(5), 444–451. https://doi.org/10.1192/bjp.175.5.444
- Biele, G., Gustavson, K., Czajkowski, N. O., Nilsen, R. M., Reichborn-Kjennerud, T., Magnus, P. M., Stoltenberg, C., & Aase, H. (2019). Bias from self selection and loss to follow-up in prospective cohort studies. *European Journal* of Epidemiology, 34(10), 927–938. https://doi.org/10.1007/ s10654-019-00550-1
- Bjørk, M., Riedel, B., Spigset, O., Veiby, G., Kolstad, E., Daltveit, A. K., & Gilhus, N. E. (2018). Association of folic acid supplementation during pregnancy with the risk of autistic traits in children exposed to antiepileptic drugs in utero. *JAMA Neurology*, 75(2), 160–168. https://doi. org/10.1001/jamaneurol.2017.3897
- Bottema-Beutel, K., Kapp, S. K., Lester, J. N., Sasson, N. J., & Hand, B. N. (2021). Avoiding Ableist Language: Suggestions for Autism Researchers. *Autism in Adulthood*, 3(1), 18–29. https://doi.org/10.1089/aut.2020.0014
- Chesnut, S. R., Wei, T., Barnard-Brak, L., & Richman, D. M. (2017). A meta-analysis of the Social Communication Questionnaire: Screening for autism spectrum disorder. *Autism*, 21(8), 920–928. https://doi. org/10.1177/1362361316660065
- Cheung, G. W., & Rensvold, R. B. (2002). Evaluating goodnessof-fit indexes for testing measurement invariance. *Structural Equation Modeling: A Multidisciplinary Journal*, 9(2), 233–255. https://doi.org/10.1207/S15328007SEM0902_5
- Evans, S. C., Boan, A. D., Bradley, C., & Carpenter, L. A. (2019). Sex/gender differences in screening for autism spectrum disorder: Implications for evidence-based assessment. *Journal of Clinical Child & Adolescent Psychology*, 48(6), 840–854. https://doi.org/10.1080/15374416.2018.1437734
- Firestein, M. R., Romeo, R. D., Winstead, H., Goldman, D. A., Grobman, W. A., Haas, D., Mercer, B., Parker, C., Parry, S., Reddy, U., Silver, R., Simhan, H., Wapner, R. J., & Champagne, F. A. (2022). Elevated prenatal maternal sex hormones, but not placental aromatase, are associated with child neurodevelopment. *Hormones and Behavior*, 140, 105125. https://doi.org/10.1016/j.yhbeh.2022.105125
- Forster, M., & Sober, E. (1994). How to tell when simpler, more unified, or less ad hoc theories will provide more accurate predictions. *The British Journal for the Philosophy of Science*, 45(1), 1–35.
- Frazier, T. W., Youngstrom, E. A., Hardan, A. Y., Georgiades, S., Constantino, J. N., & Eng, C. (2015). Quantitative autism symptom patterns recapitulate differential mechanisms of genetic transmission in single and multiple incidence families. *Molecular Autism*, 6(1), 58. https://doi.org/10.1186/ s13229-015-0050-z
- Gau, S. S.-F., Lee, C.-M., Lai, M.-C., Chiu, Y.-N., Huang, Y.-F., Kao, J.-D., & Wu, Y.-Y. (2011). Psychometric properties of the Chinese version of the Social Communication

Questionnaire. *Research in Autism Spectrum Disorders*, 5(2), 809–818. https://doi.org/10.1016/j.rasd.2010.09.010

- Grove, R., Paynter, J., Joosten, A., Vivanti, G., Dissanayake, C., & Eapen, V. (2019). Factor structure of the Social Communication Questionnaire in preschool aged autistic children. *Journal of Child and Family Studies*, 28(12), 3385–3391. https://doi.org/10.1007/s10826-019-01519-0
- Hannigan, L. J., Corfield, E. C., Askelund, A. D., Askeland, R. B., Hegemann, L., Jensen, P., Pettersen, J. H., Rayner, C., Ayorech, Z., Bakken, N. R., Wootton, R. E., & Havdahl, A. (2023, September 6). *phenotools: An R package to facilitate efficient and reproducible use of phenotypic data from MoBa and linked registry sources in the TSD environment*. https://doi.org/10.17605/OSF.IO/6G8BJ
- Havdahl, K. A., Bishop, S. L., Surén, P., Øyen, A.-S., Lord, C., Pickles, A., von Tetzchner, S., Schjølberg, S., Gunnes, N., Hornig, M., Lipkin, W. I., Susser, E., Bresnahan, M., Magnus, P., Stenberg, N., Reichborn-Kjennerud, T., & Stoltenberg, C. (2017). The influence of parental concern on the utility of autism diagnostic instruments. *Autism Research*, 10(10), 1672–1686. https://doi.org/10.1002/ aur.1817
- Hull, L., Petrides, K. V., & Mandy, W. (2020). The female autism phenotype and camouflaging: A narrative review. *Review Journal of Autism and Developmental Disorders*, 7(4), 306– 317. https://doi.org/10.1007/s40489-020-00197-9
- Jorgensen, T. D., Pornprasertmanit, S., Schoemann, A. M., & Rosseel, Y. (2022). SemTools: Useful tools for structural equation modeling. https://CRAN.R-project.org/package =semTools
- Karaminis, T., & Stavrakaki, S. (2022). The psychometric properties of the Greek version of the Social Communication Questionnaire. Autism Research: Official Journal of the International Society for Autism Research, 15(9), 1768– 1780. https://doi.org/10.1002/aur.2790
- Kidd, S. A., Berry-Kravis, E., Choo, T. H., Chen, C., Esler, A., Hoffmann, A., Andrews, H. F., & Kaufmann, W. E. (2020). Improving the diagnosis of autism spectrum disorder in fragile X syndrome by adapting the Social Communication Questionnaire and the Social Responsiveness Scale-2. *Journal of Autism and Developmental Disorders*, 50(9), 3276–3295. https://doi. org/10.1007/s10803-019-04148-0
- Kim, J. I., Lee, J., Lee, K.-S., Lee, Y. A., Shin, C. H., Hong, Y.-C., Kim, B.-N., & Lim, Y.-H. (2021). Association of phthalate exposure with autistic traits in children. *Environment International*, 157, 106775. https://doi.org/10.1016/j. envint.2021.106775
- Kline, R. B. (2016). *Principles and practice of structural equation modeling* (4th ed.). Guilford Press.
- Krakowski, A. D., Cost, K. T., Szatmari, P., Anagnostou, E., Crosbie, J., Schachar, R., Duku, E., Georgiades, S., Ayub, M., Kelley, E., Nicolson, R., Pullenayegum, E., & Barnett-Tapia, C. (2022). Characterizing the ASD–ADHD phenotype: Measurement structure and invariance in a clinical sample. *Journal of Child Psychology and Psychiatry*, 63, 1534–1543. https://doi.org/10.1111/jcpp.13609
- Lai, M.-C., Lombardo, M. V., Auyeung, B., Chakrabarti, B., & Baron-Cohen, S. (2015). Sex/gender differences and autism: Setting the scene for future research. *Journal of*

the American Academy of Child & Adolescent Psychiatry, 54(1), 11–24. https://doi.org/10.1016/j.jaac.2014.10.003

- Lai, M.-C., Lombardo, M. V., Ruigrok, A. N., Chakrabarti, B., Auyeung, B., Szatmari, P., Happé, F., & Baron-Cohen, S. (2017). Quantifying and exploring camouflaging in men and women with autism. *Autism*, 21(6), 690–702. https:// doi.org/10.1177/1362361316671012
- Liu, S.-Y., Wang, X., Chen, Q., Chen, J.-J., Jin, C.-K., Zhan, X.-L., Guo, C.-H., Li, X.-H., Lin, L.-Z., & Jing, J. (2022). The validity and reliability of the simplified Chinese version of the Social Communication Questionnaire. *Autism Research*, 15(9), 1732–1741. https://doi.org/10.1002/aur.2726
- Livingston, L. A., & Happé, F. (2017). Conceptualising compensation in neurodevelopmental disorders: Reflections from autism spectrum disorder. *Neuroscience & Biobehavioral Reviews*, 80, 729–742. https://doi.org/10.1016/j.neubiorev.2017.06.005
- Lord, C., & Bishop, S. L. (2021). Let's be clear that "autism spectrum disorder symptoms" are not always related to autism spectrum disorder. *American Journal of Psychiatry*, 178(8), 680–682. https://doi.org/10.1176/appi.ajp.2021.21060578
- Lord, C., Rutter, M., & Le Couteur, A. (1994). Autism diagnostic interview-revised: A revised version of a diagnostic interview for caregivers of individuals with possible pervasive developmental disorders. *Journal of Autism and Developmental Disorders*, 24(5), 659–685. https://doi. org/10.1007/BF02172145
- Magnus, P., Birke, C., Vejrup, K., Haugan, A., Alsaker, E., Daltveit, A. K., Handal, M., Haugen, M., Høiseth, G., Knudsen, G. P., Paltiel, L., Schreuder, P., Tambs, K., Vold, L., & Stoltenberg, C. (2016). Cohort profile update: The Norwegian Mother and Child Cohort Study (MoBa). *International Journal of Epidemiology*, 45(2), 382–388. https://doi.org/10.1093/ije/dyw029
- Magnus, P., Irgens, L. M., Haug, K., Nystad, W., Skjærven, R., Stoltenberg, C. & The Moba Study Group, (2006), Cohort profile: The Norwegian Mother and Child Cohort Study (MoBa). *International Journal of Epidemiology*, 35(5), 1146–1150. https://doi.org/10.1093/ije/dyl170
- Magyar, C. I., Pandolfi, V., & Dill, C. A. (2012). An initial evaluation of the Social Communication Questionnaire for the assessment of autism spectrum disorders in children with down syndrome. *Journal of Developmental & Behavioral Pediatrics*, 33(2), 134–145. https://doi.org/10.1097/DBP. 0b013e318240d3d9
- Mandy, W., & Lai, M.-C. (2017). Towards sex- and genderinformed autism research. *Autism*, 21(6), 643–645. https:// doi.org/10.1177/1362361317706904
- Martin, J., Hamshere, M. L., O'Donovan, M. C., Rutter, M., & Thapar, A. (2014). Factor structure of autistic traits in children with ADHD. *Journal of Autism and Developmental Disorders*, 44(1), 204–215. https://doi.org/10.1007/s10803-013-1865-0
- Meade, A. W., Johnson, E. C., & Braddy, P. W. (2008). Power and sensitivity of alternative fit indices in tests of measurement invariance. *Journal of Applied Psychology*, 93(3), 568–592. https://doi.org/10.1037/0021-9010.93.3.568
- Mulligan, A., Richardson, T., Anney, R. J. L., & Gill, M. (2009). The Social Communication Questionnaire in a sample of the general population of school-going children. *Irish*

Journal of Medical Science, 178(2), 193–199. https://doi. org/10.1007/s11845-008-0184-5

- Muthén, L. K., & Muthén, B. O. (1998–2017). *Mplus user's guide* (8th ed.).
- Piven, J., Palmer, P., Jacobi, D., Childress, D., & Arndt, S. (1997). Broader autism phenotype: Evidence from a family history study of multiple-incidence autism families. *The American Journal of Psychiatry*, 154(2), 185–190. https:// doi.org/10.1176/ajp.154.2.185
- Raiche, G., & Magis, D. (2020). NFactors: Parallel analysis and other non graphical solutions to the Cattell Scree Test (R Package Version 2.4.1). https://CRAN.R-project.org/ package=nFactors
- R Core Team. (2021). R: A language and environment for statistical computing. R Foundation for Statistical Computing. https://www.R-project.org/
- Revelle, W. (2021). Psych: Procedures for psychological, psychometric, and personality research (R Package Version 2.1.9). Northwestern University. https://CRAN.R-project. org/package=psych
- Robinson, E. B., Lichtenstein, P., Anckarsäter, H., Happé, F., & Ronald, A. (2013). Examining and interpreting the female protective effect against autistic behavior. *Proceedings of the National Academy of Sciences*, *110*(13), 5258–5262. https://doi.org/10.1073/pnas.1211070110
- Rosseel, Y. (2012). Lavaan: An R package for structural equation modeling. *Journal of Statistical Software*, 48(2). https://doi. org/10.18637/jss.v048.i02
- Rutter, M., Lord, C., & Bailey, A. (2003). *The Social Communication Questionnaire: Manual*. Western Psychological Services.
- Sasson, N. J., & Bottema-Beutel, K. (2022). Studies of autistic traits in the general population are not studies of autism. *Autism*, 26(4), 1007–1008. https://doi.org/10.1177/136236132110 58515
- Stoltenberg, C., Schjølberg, S., Bresnahan, M., Hornig, M., Hirtz, D., Dahl, C., Lie, K. K., Reichborn-Kjennerud, T., Schreuder, P., Alsaker, E., Øyen, A.-S., Magnus, P., Surén, P., Susser, E., & Lipkin, W. I. (2010). The Autism Birth Cohort: A paradigm for gene–environment–timing research. *Molecular Psychiatry*, 15(7), Article 7. https://doi. org/10.1038/mp.2009.143
- Surén, P., Havdahl, A., Øyen, A.-S., Schjølberg, S., Reichborn-Kjennerud, T., Magnus, P., Bakken, I. J. L., & Stoltenberg, C. (2019). Diagnosing autism spectrum disorder among children in Norway. *Tidsskrift for Den norske legeforening*, 139, 1–14. https://doi.org/10.4045/tidsskr. 18.0960
- Surén, P., Saasen-Havdahl, A., Bresnahan, M., Hirtz, D., Hornig, M., Lord, C., Reichborn-Kjennerud, T., Schjølberg, S., Øyen, A.-S., Magnus, P., Susser, E., Lipkin, W. I., & Stoltenberg, C. (2019). Sensitivity and specificity of early screening for autism. *BJPsych Open*, 5(3), Article e41. https://doi.org/10.1192/bjo.2019.34
- Taylor, E. C., Livingston, L. A., Clutterbuck, R. A., Shah, P., & Payne, C. (2020). Psychometric concerns with the 10-item Autism-Spectrum Quotient (AQ10) as a measure of trait autism in the general population. *Experimental Results*, 1, Article e3. https://doi.org/10.1017/exp.2019.3
- Vejrup, K., Magnus, P., & Magnus, M. (2021). Lost to followup in the Norwegian mother, father and child cohort study.

Paediatric and Perinatal Epidemiology, 36(2), 300–309. https://doi.org/10.1111/ppe.12821

- Warrier, V., Zhang, X., Reed, P., Havdahl, A., Moore, T. M., Cliquet, F., Leblond, C. S., Rolland, T., Rosengren, A., Rowitch, D. H., Hurles, M. E., Geschwind, D. H., Børglum, A. D., Robinson, E. B., Grove, J., Martin, H. C., Bourgeron, T., & Baron-Cohen, S. (2022). Genetic correlates of phenotypic heterogeneity in autism. *Nature Genetics*, 54, 1293– 1304. https://doi.org/10.1038/s41588-022-01072-5
- Wei, T., Chesnut, S. R., Barnard-Brak, L., & Richman, D. (2015). Psychometric analysis of the Social Communication Questionnaire using an item-response theory framework: Implications for the use of the lifetime and current forms. *Journal of Psychopathology and Behavioral Assessment*, 37(3), 469–480. https://doi.org/10.1007/s10862-014-9468-4
- Wigdor, E. M., Weiner, D. J., Grove, J., Fu, J. M., Thompson, W. K., Carey, C. E., Baya, N., van der Merwe, C., Walters, R. K., Satterstrom, F. K., Palmer, D. S., Rosengren, A., Bybjerg-Grauholm, J., Hougaard, D. M., Mortensen, P. B., Daly, M. J., Talkowski, M. E., Sanders, S. J., Bishop, S. L., . . .Robinson, E. B. (2022). The female protective effect against autism spectrum disorder. *Cell Genomics*, 2(6), 100134. https://doi.org/10.1016/j.xgen.2022.100134
- Wu, H., & Estabrook, R. (2016). Identification of confirmatory factor analysis models of different levels of invariance for ordered categorical outcomes. *Psychometrika*, 81(4), 1014– 1045. https://doi.org/10.1007/s11336-016-9506-0
- Yoon, M., & Lai, M. H. C. (2018). Testing factorial invariance with unbalanced samples. *Structural Equation Modeling: A Multidisciplinary Journal*, 25(2), 201–213. https://doi.org/1 0.1080/10705511.2017.1387859